Author's response to reviews

Title: Trends in dermatomyositis- and polymyositis-related mortality in the state of Sao Paulo, Brazil, 1985-2007: multiple cause-of-death analysis

Authors:

Augusto H Santo (auhsanto@usp.br)
José Maria P Souza (jmpsouza@usp.br)
Celso E Pinheiro (celso@datasus.gov.br)
Deborah CC Souza (dcolucci@hotmail.com)
Emilia I Sato (eisato@unifesp.br)

Version: 4 Date: 10 October 2010

Author's response to reviews: see over
São Paulo, October, 10 2010.

To BMC Public Health

Please find attached the revised manuscript of the paper entitled "Trends in dermatomyositis- and polymyositis-related mortality in the State of Sao Paulo, Brazil, 1985-2007: multiple cause-of-death analysis", submitted to BMC Public Health.

This letter intends to describe the changes to the format of the manuscript required by the Editorial Production Team.

From the major revisions required, the list of the authors was changed as suggested and the tables included within the text at the end of the manuscript.

From the minor revisions required, the format of the title was changed, all highlighted text was removed, and the manuscript, its references, tables and figures were checked for typographical and typing errors.

Please let me know if any problem still remains.

Respectfully,

Augusto Hasiak Santo
Rua Pará, 21 – apt 62
01243-020 – São Paulo/SP
Brazil
Telephone number: # 55 11 32584760
Fax number: # 55 11 32592612
E-mail: auhsanto@usp.br